Author’s response to reviews

Title: Is the place of birth related to the mother’s satisfaction with childbirth? A cross-sectional study in a rural district of the Lao People’s Democratic Republic (Lao PDR)

Authors:

Tomomi Takayama (tomomi-t@umin.net)
Khampheng Phongluxa (kampheng_p@hotmail.com)
Daisuke Nonaka (laodaisuke@hotmail.co.jp)
Chika Sato (sachika.ituku@gmail.com)
Ernesto Gregorio (junr.gregorio@yahoo.com)
Nouhak Inthavong (nouhak_25@yahoo.com)
Tiengkham Pongvongsa (tiengkhampvs@gmail.com)
Sengchanh Kounnavong (sengchanhkounnavong@hotmail.com)
Jun Kobayashi (junkbalao@gmail.com)

Version: 3 Date: 27 Aug 2019

Author’s response to reviews:

Author’s response letter

BMC Pregnancy and Childbirth Submission: PRCH-D-19-00553R2

Title: Is the place of birth related to the mother’s satisfaction with childbirth? A cross-sectional study in a rural district of the Lao People’s Democratic Republic (Lao PDR)

Authors:

Tomomi Takayama; corresponding author (tomomi-t@umin.net)
Khampheng Phongluxa (khampheng_p@hotmail.com)
Daisuke Nonaka (laodaisuke@hotmail.co.jp)
Author’s response to reviews

Dear Dr. Janelle Coore,

Thank you very much for the review and comments. We have revised our manuscript according to the comments. Please see a point-by-point response to the comments below.

Sincerely yours,

Tomomi Takayama

1. Please also state the role of the funding body in the design of the study, collection, analysis, interpretation of data, and in writing the manuscript, in the Funding section.

We have changed the following sentence in the Funding section (Funding section, line 429-431, page 24).

This work was partly supported by The Grant for National Center for Global Health and Medicine (NCGM) (25A-5). The NCGM did not have any role in the design, collection, analysis or interpretation of the data and in the writing of the manuscript.

2. We note that SK is missing from the authors’ contributions section, but they are included on the title page. Please ensure that all authors are included in both sections. Currently, the contributions of authors TT do not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.
We added SK’s contribution and added further clarifications for TT’s contribution. (Authors’ contributions section, line 433-439, page 25).

TT was the principal investigator and drafted the manuscript with the help of DN. TT, KP and TP supervised the field survey and contributed to the data collection. TT and DN contributed to developing the sampling strategy, analyzing data and revising the manuscript. EG contributed to the interpretation of the data and revising the manuscript. NI contributed to the acquisition of the data. CS contributed to gathering local information and acquisition of the data. SK and JK contributed to the conception of the study and interpretation of the data. All authors read and approved the final manuscript.

3. Please state in the designated input box whether the maps depicted in figure 2 are your own or taken from another source. If taken from another source please acknowledge the source in the figure legend, and if it is under copyright also state the written permission given to use and adapt it. If the above conditions are not met the image needs to be removed.

Thank you for pointing it out. We have removed the figure 2. We understood that we are not permitted to modify the original google map.

4. Please remove the response to reviewers and the tracked changes manuscript from the file inventory, as it is no longer needed.

This is noted. We removed it from the file inventory.

5. Please proofread and ensure that when you upload your revised submission that it is in the final form for publication.

Thank you for your advice. We uploaded only the final form for publication and included the responses to reviewers in the designated input box.

To reviewers

Thank you very much again for your review and advices.

Sincerely yours,

Tomomi Takayama

Doctoral course student of Department of Global Health
Graduated School of Health Sciences, University of the Ryukyus.
207 Uehara, Nishihara-cho, Nakagami-gun, Okinawa, 903-0215, Japan
Tel: +81-98-895-1286
Mail: tomomi-t@umin.net